Author's response to reviews

Title: 'Palliative care': a contradiction in terms? A qualitative study among cancer patients with a Turkish or Moroccan background, their relatives and care providers

Authors:

Fuusje M De Graaff (f.m.degraaff@uva.nl)
Anneke L Francke (a.francke@nivel.nl)
Maria ETC Van den Muijsenbergh (M.vandenMuijsenbergh@elg.umcn.nl)
Sjaak Van der Geest (S.vanderGeest@uva.nl)

Version: 7 Date: 7 September 2010

Author's response to reviews: see over
Dear Eleanor Lee,

Thank you for sending me my manuscript, corrected by your editor. They reedited the two Tables. But while comparing your version with the version I send on 22nd of July, I still found some mistakes in my text. I have corrected the manuscript by skipping the dots between the statement and indication of the quoted person in the quotes on page 16, 17, 18, 19, 20 and 26. Moreover I corrected the punctuation in the references 38, 40, 41, 42 and 48 changing a comma into a colon. And I skipped the reference to Mobeirek et al. between nr 41 and 42, because this reference is skipped in the text. Hoping the manuscript is now 100% correct, I thank you for all improvements,

Yours faithfully

Fuusje de Graaff